Author's response to reviews

Title: Bilateral ossification of the auricles: an unusual entity and review of the literature

Authors:

Nicholas S Mastronikolis (nmastr@otenet.gr)
Peter Zampakis (pzampakis@gmail.com)
Christina Kalogeropoulou (rat@med.upatras.gr)
Theodoros Stathas (tstathas@yahoo.com)
Vassiliki Siabi (vsibi@med.upatras.gr)
Eleni Geropoulou (egeropoulou@med.upatras.gr)
Panos D Goumas (pdgoumas@med.upatras.gr)

Version: 3 Date: 24 September 2009

Author's response to reviews:

Dr. Nicholas S. Mastronikolis, M.D., PhD
Lecturer
Department of Otolaryngology-Head and Neck Surgery
University Hospital of Patras
Rion, Patras 26500
Greece
Tel: +30-2610-999264
Fax: +30-2610-993986
Email: nmastr@otenet.gr

Dr. Thomas Stamm (Managing Editor-in-Cief)
University of Munster
Patras 24/09/09

Dear Dr. Stamm

thank you very much for the acceptance of our paper entitled: “Bilateral ossification of the auricles: an unusual entity and review of the literature”, in your distinguished journal. We quote below major and minor revisions according to your suggestions
Major revisions
1. Figure 5 is mentioned at the bottom of page 5.
2. All figures are cited as 'figure' on page 14.

Minor revisions
1. We have already removed the keywords section from the abstract.
2. We edit table according to your suggestions without vertical lines (page 15)